Author's response to reviews

Title: Chronic virus infections supress atopy but not asthma in a set of children from a large Latin American city

Authors:

  Rafael V Veiga (rafael.veiga@uol.com.br)
  Sérgio S Cunha (cunhatmt@hotmail.com)
  Vitor CC Dattoli (vitordattoli@gmail.com)
  Álvaro C Cruz (cruz.proar@gmail.com)
  Philip J Cooper (pcooper@ecnet.ec)
  Laura C Rodrigues (laura.Rodrigues@lshtm.ac.uk)
  Maurício L Barreto (mauricio@ufba.br)
  Neuza M Alcantara-Neves (neuza@ufba.br)

Version: 5 Date: 22 February 2011

Author's response to reviews: see over
Dear Editor,

We are re-resubmitting the manuscript: 2056709653401617, entitled “Chronic virus infections suppress atopy but not asthma in a set of children from a large Latin American city: A cross-section study”, corrected according to the Editor comments, as detailed below. I hope it is now suitable to be published by BMC pulmonary.

Sincerely,

______________________________

Neuza M. Alcantara-Neves
Instituto de Ciências da Saúde
Universidade Federal da Bahia
Avenida Reitor Miguel Calmon, s/n, Canela
CEP – 40110-100. Salvador, Bahia, Brazil
E-mail: neuzalcantara@gmail.com
<table>
<thead>
<tr>
<th>Editor comments</th>
<th>Answers</th>
</tr>
</thead>
<tbody>
<tr>
<td>Please include a 'Competing interests' section between the Conclusions and Authors' contributions. “If there are none to declare, please write 'The authors declare that they have no competing interests”.”</td>
<td>These were done, please see lines 329- 330.</td>
</tr>
<tr>
<td>“Please include an Authors' contributions section before the Acknowledgements and Reference list”.</td>
<td>This was done, please see lines 332- 338.</td>
</tr>
<tr>
<td>“We strongly encourage you to include an Acknowledgements section between the Authors’ contributions section and Reference list”. “Please list the source(s) of funding for the study, for each author, and for the manuscript preparation in the acknowledgements section”.</td>
<td>These were done, please see lines 340-344.</td>
</tr>
<tr>
<td>“Authors must describe the role of the funding body, if any, in study design; in the collection, analysis, and interpretation of data; in the writing of the manuscript; and in the decision to submit the manuscript for publication.”</td>
<td>This was done, please see lines 344-345.</td>
</tr>
<tr>
<td>We would be grateful if you could address the comments in a revised manuscript and provide a cover letter giving a point-by-point response to the concerns.</td>
<td>This was done.</td>
</tr>
<tr>
<td>Please also highlight (with 'tracked changes'/coloured/underlines/highlighted text) all changes made when revising the manuscript to make it easier for the Editors to give you a prompt decision on your manuscript.</td>
<td>This was done.</td>
</tr>
<tr>
<td>Please also ensure that your revised manuscript conforms to the journal style (<a href="http://www.biomedcentral.com/info/ifora/medicine_journals">http://www.biomedcentral.com/info/ifora/medicine_journals</a> ). It is important that your files are correctly formatted.</td>
<td>This was done.</td>
</tr>
</tbody>
</table>